Author's response to reviews

Title: Epidemiology of episodic adenolymphangitis: a longitudinal prospective surveillance among a rural community endemic for bancroftian filariasis in coastal Orissa, India

Authors:

Bontha V Babu (babubv@satyam.net.in)
Abhay N Nayak (bontha99@hotmail.com)
Kalpataru Dhal (bontha99@hotmail.com)

Version: 5 Date: 16 May 2005

Author’s response to reviews:

16th May 2005

To
The Editor
BMC Public Health

Sub.: Ms: 1066802644484799 - Submission of FINAL manuscript for consideration of publication in the BMC Public Health

Sir,

Thank you very much for your letter and considering our manuscript for acceptance for publishing in BMC Public Health. I could not send the final version of the manuscript immediately after receiving your mail, as I was in Geneva till 14th May.

I am herewith provided the point-to-point information to indicate where and how the formatting has been made.

1. The changes have been in the list of authors to make identical with that given in submission system.
2. Affiliations of authors are given as suggested.
3. Title and authors’ details in Page 2 are deleted.
4. Abstract is modified as suggested. The revised abstract has been uploaded in the submission page.
5. 'Description of each author's role' is renamed as 'Authors' contributions'.
6. Figure legend has been placed before tables.

In addition, the manuscript has been rechecked thoroughly for grammatical errors and correctness of references.

I request you to consider this manuscript for publication, and please let me know if I need to do further.

With regards,

Sincerely

B. V. Babu